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	CRUISE: EX10-04 Leg II
Bitung to Bitung
	PERSONNEL ON BOARD:

NOAA OER:

1. Mashkoor Malik (ERT Inc)

2. Dave Lovalvo (Eastern Oceanics)

3. Webb Pinner (2020 LLC)

4. Jeremy Potter (OER)

Others:

5. Doug Jongeward (PMEL)

6. John Sherrin (UVic)

7. Xerandy (BPPT)

8. Major Muddan (TNI)

9. Dave Wright (UCAR/JOSS)

10. Joel DeMello (UCAR/JOSS)

11. Tom Kok (UCAR/JOSS)

12. Tom Pierce (UCAR/JOSS)

13. Karl McLetchie (UCAR/JOSS)

14. Brian Bingham (UCAR/JOSS)
15. Randy Prickett (UCAR/JOSS)

16. Eric Prechtl (UCAR/JOSS)

17. Roland Brian (UCAR/JOSS)

18. Joe Biscotti (UCAR/JOSS)

	DATE/TIME FILED:  06/30/2010 1109 UTC

FILED BY: Jeremy Potter
	

	VESSEL:  NOAA Ship Okeanos Explorer (EX)


	

	GEOGRAPHIC AREA:  Operating north of North Sulawesi, Indonesia 
	


=============================================================================

SUMMARY:    It was another big day for the Okeanos Explorer Program. The shore-side and ship-board team completed a fantastic first ROV dive. From both the operations and science perspective, it was a great day.

Onboard operations on 6/29 included multibeam data acquisition, our first ROV dive, and three CTD casts. Jakarta and Seattle ECCs were fully staffed during the ROV dive. 

NOTE: As of today, local ship time is 8 hrs ahead of UTC.  Jakarta is 7 hrs ahead of UTC. East Coast is 4 hrs behind UTC. West Coast is 7 hrs behind UTC.

============================================================================
ROV:

The first ROV dive of the expedition took place today. The landing point was at a depth of 2,350 m, approximately 450 m below the NW end of the summit crest. The ROV proceeded in an easterly direction until it reached the NW crest. Towards the summit there was an increase in the amount of fresh pyroclastic sediment and a large number of bivalves. A dense hydrothermal plume was observed along the middle portion of the crest at the end of the dive but the seafloor source was not found. Details of the dive are available in the iChat event log. Based on these findings, a set of CTD casts were planned from the south of the crest to the north of the crest to determine whether venting was occurring on the northern or southern side of the crest.
The LinkQuest USBL transceiver which is mounted on the bottom of the ship in the tranducer fairing appears to be unable to transmit a strong enough signal to interrogate the camera platform transponder (e.g., we cannot track the camera sled). So far all troubleshooting seems to indicate the problem is in the transmitter portion of the transceiver. As a result, we have minimal ability to fully troubleshoot and/or repair because the installation is inaccessible by divers.  
SURVEY/CTD:
Multibeam mapping operations continued until soon before the ROV dive began.  The Science Team selected three locations for CTD casts:
CTD 1: 4° 40 20.5 N 125° 5 12.0 E, measured depth > 1,950 m (S of summit)

CTD 2: 4° 40.56    N 125° 5.23    E, measured depth = 1,880 m (summit crest)

CTD 3: 4° 40 40.3 N 125° 5 14.1 E, measured depth >1,950 m (N. of summit)

Preliminary data and information from the casts included evidence of hydrothermal activity. 
SHIP NETWORK

Low-speed network – Performance has improved compare to the beginning of EX-10-04 Leg1 but continues to be poorer than EX-10-02.  The cause of the poor performance is perhaps related to an unknown source of inbound traffic.  OMAO shore-side support personnel are aware of the issue and will continue working to identify the cause and solution on Tuesday local time. 
High-speed Network – On June 20, 2010 bandwidth to the high-speed link was increased from 10Mbps to 20 Mbps per the NOAA contract with MTN. OER pays $6K per day for the service to increase the size and quality of the video feeds to the ECCs.  When the bandwidth was expanded, the ECCs experienced a serious degradation in video quality and internet access through the visitor’s network was completely lost.  Once the total size of the all three feeds was reduced to less than 10Mbps the video quality reverted back to previous levels and internet access for the visitors returned.  The ETs have confirmed with MTN that the link to the Okeanos Explorer is 20Mbps and the OMAO network engineer has confirmed the system should support outbound traffic up to 20 Mbps.  The problem appears to be linked to the ship’s TurboIP box.  The TurboIP box is responsible for accelerating TCP-based traffic (i.e web, ftp) over the satellite link. With the TurboIP box removed the ship is able to transmit video up to 20Mbps but ftp transfers fail.  MTN is contacting the manufacturer for additional assistance.  Though OMAO is responsible for solving the problem, OER is responsible for paying all MTN contract costs. To date, OER has spent $54K for services that it is unable to use because of the unknown problem. 
OMAO and OER discussions on network improvements continue. The Chief ET ceased trying to keep hours that overlap with Silver Spring personnel. Communication and testing across each group continues to be difficult due to varying time zones. OMAO and OER are arranging for a contract network engineer to examine and trouble-shoot the network from shore. 

VSAT

MTN and OMAO are looking into the >10MBps bandwidth issues described above.

Dynamic Positioning

DP system worked quite well during ROV operations. There are no issues to report.
OTHER:

A replacement part is on order to remedy the problem with the interface to the satellite phone. The Iridium phone can only be operated from the bridge and Inmarsat can only be used from the ET shop until the part is replaced during the next in-port.
Mission Video Engineers working in collaboration with Chief ET Conway greatly reduced interference in the ship entertainment system. 
============================================================================================================================================================
PLAN OF THE DAY

Wednesday, June 30, 2010

OPERATIONS: Exploration Indonesia

	0000
	Underway as before
Conducting mapping operations

	0700
	Secure all overboard discharges

	0730
	Bow and Stern thrusters online

	0745
	Safety Brief (Bridge)

	0815
	Commence ROV deployment

	1300-1500
	Employee Relations Meeting
EEO Meeting

Safety Management Meeting

Supervisory Management Meeting

	1500
	Ops Brief (Forward Lounge)

	1700
	ROV on deck
Resume mapping operations and possibly CTD ops


· Conduct ROV dive over Kawio Barat in DP

· Continue multibeam operations
· 0030 conduct XBT
· Conduct monthly meetings in the forward lounge
On Deck for Thursday:

· Conduct ROV operations

· Possibly conduct CTD ops
· Conduct multibeam and XBT operations overnight
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